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CS1 CS1 ObservationsObservations

17/4: Pulsar tracking experiment17/4: Pulsar tracking experiment
6 Tiles6 Tiles
Tracking did not workTracking did not work

Under investigationUnder investigation

B0329+54_20080417B0329+54_20080417

18/4 18/4 –– 20/4: HBA Milky Way experiment20/4: HBA Milky Way experiment
4 Tiles + 20 dipoles; CS001 RCU 8&9 not ok4 Tiles + 20 dipoles; CS001 RCU 8&9 not ok
Test:Test:

24 24 microstationmicrostation still operationalstill operational
Mapping galactic structure around Cygnus at 110Mapping galactic structure around Cygnus at 110--190 MHz190 MHz
Rain?Rain?

20082008--06100 06100 
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CS1 StatusCS1 Status

StationsStations
CS001 RCU 8 and 9 single element HBA1  no signalCS001 RCU 8 and 9 single element HBA1  no signal

Connector probably broken from print (weak connection)Connector probably broken from print (weak connection)
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CS1 StatusCS1 Status

NetworkNetwork
LightpathLightpath LOFAR LOFAR –– AmsterdamAmsterdam

Gateway installed GroningenGateway installed Groningen
Tomorrow Tomorrow ListFENListFEN connected to the connected to the gatwaygatway
Next: connection with SARA GRID storageNext: connection with SARA GRID storage
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CS1 StatusCS1 Status

CEPCEP
Problem storing data on list001&list002Problem storing data on list001&list002

Tested & working; cause earlier problem unknownTested & working; cause earlier problem unknown

Online storage cleaned upOnline storage cleaned up
64 bit offline nodes64 bit offline nodes

InstallationInstallation
Connect to Connect to lifslifs InfinibandInfiniband networknetwork
Planned for tomorrowPlanned for tomorrow

Distribute datasets over storage nodesDistribute datasets over storage nodes
Shell script replaced by Python script offering more flexibilityShell script replaced by Python script offering more flexibility
Being testedBeing tested

CEP startup still supported by CEP teamCEP startup still supported by CEP team
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CS1 PlanningCS1 Planning

Tuesday & Wednesday (today) TNO/Tuesday & Wednesday (today) TNO/TTüüVV teststests
Thursday 24/4 LBA test observationThursday 24/4 LBA test observation
Weekend LBA observation?Weekend LBA observation?


